Author's response to reviews

Title: Treatment of Myofascial Trigger Points in Patients with Chronic Shoulder Pain: A Randomized Controlled Trial

Authors:

Carel Bron (carelbron@mac.com)
Arthur de Gast (adgast@diakhuis.nl)
Jan Dommerholt (dommerholt@bethesdaphysiocare.com)
Boudewijn Stegenga (b.stegenga@kchir.umcg.nl)
Michel Wensing (m.wensing@iq.umcn.nl)
Rob A.B. Oostendorp (r.oostendorp@iq.umcn.nl)

Version: 3 Date: 8 December 2010

Author's response to reviews: see over
December 6, 2010

Editor-in-chief
BMC Medicine

We would like to submit the revised version of the manuscript “Treatment of Myofascial Trigger Points in Patients with Chronic Shoulder Pain; A Randomized Controlled Trial” for publication in BMC Medicine. We are very pleased to know that you have accepted the manuscript for publication. The requested changes in the manuscript have been highlighted in red. Our responses to the comments have been included below in this cover letter. We ensure BMC Medicine that permission is granted to the first author by Lippincott Williams & Wilkins to use up to ten images in any single work. We have consent from both subjects on all photographs to publish the pictures in the manuscript. We have attached the photography consent forms and the lifeART license agreement below in this cover letter.

Reviewer's report (3): I have read the response of the authors and all previous comments made by the reviewers. I feel that the authors have made a thorough revision of the manuscript and have answered all the major issues raised in the first round of revision. I appreciate the effort made in adding the clinical details and precise description of the interventions. The only thing I would raise at this point is that the author should describe in the methods section how the co-interventions were assessed (interview, personal diary?).

We have added a few lines at the end of the method section about the assessment of the co-interventions.

Yours sincerely,

Carel Bron, PT, MPT (PhD student)

Co-Authors: Arthur de Gast, MD, PhD; Jan Dommerholt PT, DPT; Boudewijn Stegenga DDS, PhD; Michel Wensing PhD Habil; Rob Oostendorp PT, MPT, PhD.
Life:ART License Agreement

IMPORTANT! THIS IS A LEGAL AGREEMENT BETWEEN YOU AND WOLTERS KLUWER HEALTH, INC., LIPPINCOTT WILLIAMS & WILKINS. READ THIS AGREEMENT IN ITS ENTIRETY BEFORE YOU DOWNLOAD ANY IMAGE. BY DOWNLOADING ANY IMAGE YOU AGREE TO BE BOUND BY THE TERMS OF THIS AGREEMENT.

GRANT OF LICENSE:
Wolters Kluwer Health, Inc.-Lippincott Williams & Wilkins grants you a personal, nontransferable, nonexclusive, nonsub licensable right to use the Life:ART/MediClip image(s) you have selected to download onto your computer hard drive (or other storage device) and to use the Life:ART/MediClip image and any derivatives or copies (collectively, the "Image") on one computer and with one user at a time. The publisher, Wolters Kluwer Health, Inc.-Lippincott Williams & Wilkins, retains ownership of the Image(s) as well as any subsequent copies on any media or in any form. Your License to the Life:ART/MediClip Image(s) grants you, the purchaser, permission to use, modify, and include Life:ART/MediClip Image(s) in other documents and materials, as part of a total graphic design, for yourself, your client, or your employer, subject to the following limitations:

FOR EDUCATIONAL USE:

- a credit reading, "Life:ART (and/or) MediClip Image copyright (appropriate year) Wolters Kluwer Health, Inc.-Lippincott Williams & Wilkins. All rights reserved." should appear adjacent to the Image(s) or in an obvious credits page or screen;

- you may incorporate up to 10 Life:ART/MediClip Images in any single work from any one collection. To incorporate more than 10 Life:ART/MediClip Images in any single work, please contact our Licensing Department at licensing@wolterskluwer.com. Additional licensing fees may be applied.

- the Image(s) may not be distributed, sublicensed or made available for use or distribution as art for reproduction (clipart), any form of stock design, prints, negatives, positives, mats, engravings, etc., without written permission from Wolters Kluwer Health, Inc.-Lippincott Williams & Wilkins; and no rights to the Image(s) may be granted other than the right to use the Image(s) as part of the product made by you;

- Image(s) used in electronic form, including multimedia applications and Internet or intranet sites, should be for viewing purposes only, i.e., prepared as embedded or inline graphics, and cannot be in a downloadable or FIP format;

- pornographic, defamatory, libelous or otherwise unlawful use of the Image is prohibited.

FOR RESALES PURPOSES:

- you may make the Image available to ten (10) separate individuals (cumulatively over the Term) for the sole purpose of manipulating or otherwise using the Image to create
PHOTOGRAPHIC CONSENT FORM/MODEL RELEASE

I, Mr. B.G.M. de Valk, hereby grant permission to Praktijk voor Nek-, Schouder- en Armklachten / Myofasciale pijn seminars Groningen, Paulus Potterstraat 46, 9718 TK Groningen, the Netherlands, its employees or representatives, to take and use digital images of me for use in promotional or educational materials, scientific publications, book chapters, and other medium related to the nature of the business the company engages in. These materials might include printed or electronic publications, websites or other electronic communications. I authorize the use of these images indefinitely without compensation to me. All negatives, positives, prints, digital reproductions and videotape shall be the property of Praktijk voor Nek-, Schouder- en Armklachten / Myofasciale pijn seminars Groningen, Paulus Potterstraat 46, 9718 TK Groningen, the Netherlands.

Date: December 6, 2010

Signature of adult subject: 

Full name of adult subject: B.G.M. de Valk

Address: Oude Paardendijk 4

City, State, Zip: Peize, DR, 9321 XT

Signature Witness: 

Full name of Witness: L. Bijlsma
PHOTOGRAPHY CONSENT FORM/MODEL RELEASE

I, Mrs. B.T.M. Beersma, hereby grant permission to Praktijk voor Nek-, Schouder- en Armlaenen / Myofasciale pijn seminars Groningen, Paulus Potterstraat 46, 9718 TK Groningen, the Netherlands, its employees or representatives, to take and use digital images of me for use in promotional or educational materials, scientific publications, book chapters, and other medium related to the nature of the business the company engages in. These materials might include printed or electronic publications, websites or other electronic communications. I authorize the use of these images indefinitely without compensation to me. All negatives, positives, prints, digital reproductions and videotape shall be the property of Praktijk voor Nek-, Schouder- en Armlaenen / Myofasciale pijn seminars Groningen, Paulus Potterstraat 46, 9718 TK Groningen, the Netherlands.

Date: December 6, 2010

Signature of adult subject: [Signature]

Full name of adult subject: B.T.M. Beersma

Address: Zonland 154

City, State, Zip: Groningen, GN, 9724 BR

Signature Witness: [Signature]

Full name of Witness: L. Bijlsma